Author's response to reviews

Title: Early experiences on the feasibility, acceptability, and use of malaria rapid diagnostic tests at peripheral health centres in Uganda--insights into some barriers and facilitators

Authors:

Caroline Asiimwe (caroline.asiimwe@finddiagnostics.org)
Daniel J Kyabayinze (d.kyabayinze@malariaconsortium.org)
Zephania Kyalisiima (zkyalisima@ss.mak.ac.ug)
Jane Nabakooza (jinksuganda@yahoo.com)
Moses Bajabaite (cmbaja@yahoo.com)
Helen Counihan (h.counihan@malariaconsortium.org)
James K Tibenderana (j.tibenderana@malariaconsortium.org)

Version: 11 Date: 18 January 2012

Author's response to reviews:

Dear editorial team,

We received and read through an edited version of the draft manuscript we submitted on 2 January 2012. We are happy with most of the edits made by your team. We have addressed all the major revisions recommended, including updating the title, abstract and main body of the manuscript in the system.

We have confirmed in the manuscript that all authors read and approved the revisions in the manuscript.

We have also updated the references, whose numerical order had changed slightly as a result of the edits.

Your recommendation that we crop the 2 figures has also been hided to. Figure 1 and 2 have been cropped to minimize white space around the images.

We have taken this opportunity to check the manuscript for typographical errors and we confirm that we have corrected the few existing ones. These are tracked as changes in the main body.

We have removed the additional files form the system as per your request, since they do not feature in the main body of the manuscript.

Thank you for the support rendered, in revising this manuscript

Asiimwe Caroline